Author's response to reviews

Title: Hepatic Artery Aneurysm Repair: Report of an Unusual Case

Authors:

S S Jaunoo (ssj@doctors.org.uk)
T Tang (ttang@hotmail.com)
C Uzoigwe (cw@hotmail.com)
S Walsh (swalsh@yahoo.co.uk)
M Gaunt (mgaunt@yahoo.co.uk)

Version: 5 Date: 22 February 2008

Author's response to reviews:

Many thanks for your reviewers' comments. I have made the following amendments and changes:

1. I have changed the order of the authors
2. I have reduced the number of references to 8 as requested
3. I have shortened the conclusion by removing the last sentence.
4. I have changed "America" to "American"
5. I have not included the actual incidence as there is no universally agreed value. These range from 0.25% to 0.75% but the incidence is continually rising as more and more HAA are becoming detected. Hence, after discussion with my fellow authors we feel that any quoted incidence at this time would very quickly become out-of-date.

Please let me know if any further changes are needed.

With kind regards.

SS Jaunoo